Author's response to reviews

Title: Bipolar patient treated with lithium, haloperidol and benztropine during pregnancy presented with unexpected depletion in plasma choline and phosphatidylcholine: a case report

Authors:

Maxine Gossell-Williams (maxine.gossell@uwimona.edu.jm)
Horace Fletcher (horace.fletcher@uwimona.edu.jm)
Steven Zeisel (steven_zeisel@unc.edu)

Version: 2 Date: 17 October 2007

Author's response to reviews:

Dear Editor, I am now submitting the revised paper. All the concerned expressed have been addressed.

1. Title has been change
2. Discussion has been added
3. References have been reduced
4. Grammar has been corrected
5. Web link has been placed in references 6. Competing interest has been added